HEST - 12 Nov 2019



- Exchange of the vacuum chambers in TH2: material acquisition done, still targeting January installation.
- Particle counter for miniCBM chamber in January.

Fluorescent screen for miniCBM – full integration to CUPID still

pending (software not ready).

 Rubbish successfully removed from NE8 (many thanks to Christoph and Beata).

 Ongoing review of HTP optics in preparation for BI experiment.

